Author's response to reviews

Title: Organ culture: a new model for vascular endothelium dysfunction

Authors:

Rikard Alm (rikard.alm@med.lu.se)
Lars Edvinsson (lars.edvinsson@med.lu.se)
Malin Malmsjo (malin.malmsjo@med.lu.se)

Version: 3 Date: 26 Apr 2002

Dear Editor,

Thank you for accepting this manuscript. All queries have been taken care of as follows:

- "competing interests" has been added after the acknowledgements.
- The figures do not feature the legend text “figure 1” etc.
- The figures should be cropped to minimise the white space around the image. Would you please assist in doing this. The figures are sent to you in PC office word2000 format.

The 2 feb. 2002 Lunds universitet signed a membership in the BioMed Central. Since all authors of this manuscript work at the University of Lund we should not pay the fee of 500 USD.

I hope that everything is ready for publication now.

Yours sincerely,

Malin Malmsjo, MD, PhD